Author's response to reviews

Title: Careers in ecstasy use: do ecstasy users cease of their own accord? Implications for intervention development

Authors:

Gjalt-Jorn Y Peters (gj.peters@psychology.unimaas.nl)
Gerjo Kok (g.kok@psychology.unimaas.nl)
Herman P Schaalma (herman.schaalma@psychology.unimaas.nl)

Version: 5 Date: 16 October 2008

Author's response to reviews:

Dear Mark,

to clarify the manuscripts ethical position, we have added the following sentence to the methods section:

Permission to perform this investigation was granted by the Ethical Committee Psychology of Maastricht University (the ECP).

We hope that this is what you had in mind.

Kind regards,

Gjalt-Jorn Peters